Massachusetts Institute of Technology Handout 16 6.915: Network and Computer Security October 12, 1995 Professor Ronald L. Rivest

From the NeW York Times

Attached you will find an article that appeared on yesterday’s New York Times [1], and the posting on the Cypherpunks mailing list that prompted the NYT article [2].

References

[1] J. Markoff, Discovery of Internet Flaws is Setback for On–Line Trade, New York Times, Wednesday Ocotber 11, 1995

[2] E. Brewer, P. Gauthier, I. Goldberg, D. Wagner, Basic Flaws in Internet Security and Commerce, Cypherpunks Mailing List Posting on Monday October 9 1995.